Contents lists available at ScienceDirect

# European Journal of Operational Research

journal homepage: www.elsevier.com/locate/ejor



# Emerging trends and new frontiers in community operational research



Michael P. Johnson a,\*, Gerald Midgley b,c,d,e,f, George Chichirau a

- <sup>a</sup> Department of Public Policy and Public Affairs, University of Massachusetts Boston, USA
- <sup>b</sup> Centre for Systems Studies, Business School, University of Hull, Hull, UK
- <sup>c</sup> Victoria Business School, Victoria University of Wellington, Wellington, New Zealand
- <sup>d</sup> School of Innovation, Design and Engineering, Mälardalen University, Eskilstuna, Sweden
- <sup>e</sup> School of Political and Social Sciences, University of Canterbury, Christchurch, New Zealand
- <sup>f</sup>School of Agriculture and Food Sciences, University of Queensland, Brisbane, Queensland, Australia

#### ARTICLE INFO

#### Article history: Received 4 January 2017 Accepted 15 November 2017 Available online 22 November 2017

Community operational research Community-based operations research Analytics Developing countries Systems thinking

# ABSTRACT

Community Operational Research (Community OR), and its disciplinary relation, Community-Based Operations Research, has an increasingly high profile within multiple domains that benefit from empirical and analytical approaches to problem solving. These domains are primarily concentrated within nonprofit services and local development. However, there are many other disciplines and application areas for which novel applications and extensions of Community OR could generate valuable insights. This paper identifies a number of these, distinguishing between 'emerging trends' (mostly in well-studied areas of operational research, management science and analytics) and 'new frontiers', which can be found in traditions not commonly oriented towards empirical and analytical methods for problem solving, where community-engaged decision modeling represents new ways of generating knowledge, policies and prescriptions. This paper will show how the exploration of emerging trends and new frontiers in Community OR can provide a basis for the development of innovative research agendas that can broaden the scope and impact of the decision sciences.

> © 2018 The Authors. Published by Elsevier B.V. This is an open access article under the CC BY-NC-ND license. (http://creativecommons.org/licenses/by-nc-nd/4.0/)

#### 1. Introduction

Community Operational Research (Community OR), and its disciplinary relation, Community-Based Operations Research (CBOR), has an increasingly high profile within multiple domains that benefit from empirical and analytical approaches to problem solving. Many of these areas are concentrated in human services, community and economic development, education and other non-profit services, and the nature of inquiry tends to be influenced by action research and systems thinking as much as traditional decision modeling. However, there are many other areas of inquiry in which Community OR has had only a modest presence to date.

The goal of this paper is to explain how Community OR can help identify problem opportunities, novel analytical methods, theory-building and contributions to practice in a variety of domains, some closely identified with operational research, management science and analytics (henceforth referred to generally as the 'decision sciences'), and others more firmly rooted in disciplinary

E-mail addresses: michael.johnson@umb.edu (M.P. Johnson), g.r.midgley@hull.ac.uk (G. Midgley), George.Chichirau001@umb.edu (G. Chichirau). traditions not conventionally associated with decision science. By doing so, we hope to provide encouragement and resources for researchers and practitioners who seek new applications for Community OR that support frequently-pursued Community OR values, such as community engagement, equity and social justice.

We start by providing a short introduction to Community OR, and in so doing we clarify distinctions between this and other areas of decision science. Elsewhere (Midgley, Johnson, & Chichirau, 2018), we argue that the definitive feature of Community OR is "the meaningful engagement of communities", which leaves open questions about what counts as 'meaningful' (see Ufua, Papadopoulos, & Midgley, 2018, for a discussion of this) and what constitutes a 'community' (e.g. Midgley & Ochoa-Arias, 1999). Importantly, defining Community OR in this way draws out a principle of practice (meaningful community engagement) that is present in all previously published examples of Community OR, so this is not an imposition on our discipline. However, it also proposes a normative standard for future practice and publications, to limit the possibility of 'drift' into less community-engaged forms of OR.

Note that a consequence of this definition is that there are overlaps between Community OR and other well established traditions, such as public sector OR and even OR in the private sec-

https://doi.org/10.1016/j.ejor.2017.11.032

0377-2217/© 2018 The Authors. Published by Elsevier B.V. This is an open access article under the CC BY-NC-ND license. (http://creativecommons.org/licenses/by-nc-nd/4.0/)

<sup>\*</sup> Corresponding author.

tor (see Midgley et al., 2018, for examples). For instance, perfectly good public sector OR can be client-engaged, but not community-engaged. However, when the community actually has meaningful input, the project may be both public sector and Community OR. Below, we use public sector OR as an example to compare Community OR with, although we should provide a caveat here: many of the things we discuss below are general characteristics of Community OR compared with public sector OR, and the term 'general characteristics' refers to things that are commonly found in applications. This is different from saying they are defining features of Community OR. In our view, only the 'meaningful engagement of communities' can be considered definitional (in addition, of course, to things that are definitional of all forms of OR, like modeling).

Public sector OR has traditionally addressed three areas of decision modeling impacts: efficiency, effectiveness and equity (Savas, 1969, 1978). Bardach and Patashnik (2016) express efficiency as maximizing the sum of individual utilities, and Stokey and Zeckhauser (1978) characterize efficient solutions as lying on a Pareto frontier of possible allocations of goods and services among members of a population. Effectiveness, in contrast to efficiency, seeks to identify policies or interventions that best achieve socially desirable outcomes, especially when markets diverge greatly from the common neoclassical assumption of perfect competition, or when there are no easily identifiable markets for the goods or services of interest (LeClerc, McLay, & Mayorga, 2012). Finally, notions of equity, fairness or justice address concerns that a just society may take steps to ensure that certain groups receive benefits from policies or interventions roughly commensurate to their needs (LeClerc et al., 2012). Commonly used introductions to management science, such as Winston and Albright (2016), tend not to address efficiency directly, rather concentrating on objectives most salient to private sector operations, such as minimizing cost or maximizing yield or profit. Mainstream introductions to management science are equally silent on issues related to effectiveness and equity. Even standard reviews of public sector OR (see e.g. Pollock et al., 1994) have relatively little to say about issues of equity and social justice as compared with more traditional emphases on technical modeling.

Community OR is distinguished from client-but-notcommunity-engaged public sector operational research in a number of ways. First, Community OR places great emphasis on intervention, or "purposeful action by an agent to create change" (Midgley, 2000, p.9), as opposed to observational science alone or methodological innovations outside the context of interventions. Effective Community OR interventions require a deep understanding of the problem context, a commitment to empiricism, engagements with stakeholders, and primary data collection to reflect the lived experiences of those who are engaged with the problem to be solved (e.g. Friend, 2004). Many public sector projects also involve interventions, but the majority of the publications discussing them are framed in terms of novel modeling techniques and the findings from data analyses, with the engagement of clients and stakeholders that is required for effective intervention taking a back seat. Publications about Community OR projects, in contrast, tend to emphasize the latter alongside the reporting of methodological innovations (e.g. Johnson 2012a; Midgley & Ochoa-Arias, 2004a; Ritchie et al., 1994).

Next, in line with the focus on intervention, Community OR generally has a central concern for *local engagement and impact*. This arises from a belief that many problems of greatest immediate concern to citizens (such as education, crime, housing and economic development) have a local character, and that giving local residents a say in problem identification, formulation, solution and the implementation of new prescriptions or guidelines may result in significant and/or rapid improvements in (perceived) quality of life

Community OR also usually has a concern for disadvantaged, underrepresented and underserved populations. This is about social justice, which involves efforts to promote "....a just society by challenging injustice and valuing diversity" (National Federation for Catholic Youth Ministry 2008) or ensuring "equal access to liberties, rights, and opportunities, as well as taking care of the least advantaged members of society" (Robinson, 2016). Social justice was a concern of those who first founded Community OR (e.g. Jackson, 1987; Rosenhead, 1986).

Community OR focuses on problem solving processes as well as outcomes (Midgley and Ochoa-Arias 2004b); in particular, designing interventions that are intended to improve the understanding of decision opportunities, data and solutions as much as to produce specific prescriptions or strategies (e.g. Ritchie, 2004). Johnson (2012b) argues that Community-Based Operations Research (CBOR), a domain closely aligned with Community OR, derives a great deal of value from

- Identifying problems which may not, at first glance, appear amenable to conventional OR methods;
- Formulating those problems in such a way as to prioritize diverse conceptions of values, evidence of beneficial social impacts and equity;
- Solving them (or addressing/managing them when no immediate resolution is possible) through multiple research frameworks and analytical methods that yield understanding as much as prescriptions; and
- Implementing solutions to enable capacity building and social change, with theory development being a possible outcome too.

Community OR embodies a critical approach and a concern for ethics. By 'critical approach', we mean a desire to

- Interrogate assumptions about whose conceptions of a problem count (e.g. Foote et al., 2007; Midgley & Pinzón, 2011; Ulrich, 1987, 1994, 1996);
- Explore the implications of power relationships between 'experts' who address problems, 'clients' who present problems to be solved, and communities who are the potential beneficiaries or victims of new policies or prescriptions (e.g. Córdoba & Midgley, 2006; Midgley & Milne, 1995);
- Understand the inevitably non-neutral role of the practitioner, and perceptions among stakeholders of his/her identity, which bring the need to link self-reflection with stakeholder dialogue, empirical-analytic inquiry and ideology critique (Gregory, 1992, 2000; Midgley, 1995); and
- Take seriously alternative research philosophies and methodologies, such as post-positivism, constructivism, transformative research, emancipatory inquiry and pragmatism (e.g. Creswell, 2014; Jackson, 1985; Metcalfe, 2008; Midgley, 2004; Ormerod, 2006; Taket & White, 1993).

'Ethics' refers to

- Concerns about the probity of engagements by researchers (e.g. issues of independence and honesty when there is a fee paying client and other stakeholders may suffer);
- The integrity of relationships between researchers, clients and participants, so exploitation of various kinds is avoided; and
- The consequences of decisions on those affected but not involved (Córdoba, 2009; Ormerod & Ulrich, 2013; Taket, 1994; Ulrich, 1994; Wenstop & Koppang, 2009).

Moral inquiry can shed light on the possible rights and responsibilities of stakeholders, especially in problematic situations (Mingers, 2011a). Likewise, Midgley, Munlo, and Brown (1998) follow Ulrich (1987, 1994, 1996) in arguing that every boundary judgment made in a Community OR project (about whose views and what issues to include, exclude or marginalize) is also

an ethical judgment, so ethics has to be a central concern for practice.

In contrast with many others working in operational research, management science and analytics, Community OR practitioners tend to exhibit a methodological preference for *qualitative* (e.g. Mingers & Rosenhead, 2004; Rosenhead & Mingers, 2001) and *mixed method* (e.g. Mingers and Gill, 1997) approaches to problem solving, as opposed to the traditional foci on quantitative data, mathematical modeling and the manipulation of quantitative data via algorithms. The tension between those advocating for qualitative versus quantitative methods is long-standing within OR/MS (Kirby, 2007; Mingers, 2011b, 2011c) and, to some extent at least, mixed method approaches are able to transcend this by accepting the validity and utility of both (e.g. Flood and Jackson 1991; Flood and Romm 1996; Midgley, 1992, 1996, 2000; Midgley, Nicholson, & Brennan, 2017; Mingers and Gill 1997).

Finally, Community OR practitioners generally seek to design interventions that result in community empowerment and social change. They mostly eschew the idea that problem-orientated inquiry can be 'neutral' or 'value-free' (e.g. Alrøe, 2000; Midgley, 2000; Ulrich, 1994). This empowerment and social change orientation was partly introduced as a reaction to the right wing politics of the Thatcher era in the UK (e.g. Rosenhead, 1986), but has since expanded into a broader philosophy of 'engaged OR' (Midgley et al., 2018) that provides a counterweight to both unfettered capitalism and centralized bureaucratic planning (Jackson, 1987). As such, it represents a re-imagining of what operational research can do with and for communities in general, and disadvantaged and marginalized communities in particular, often using methods not considered within the mainstream of OR in the USA (Jackson, 1988; Midgley et al., 2018). Compare Simchi-Levi's (2006, 2009) defense of narrow boundaries for OR with Ackermann et al's (2009) appeal to take Soft OR, and by extension Community OR, seriously within the discipline.

Community operational research thus has many features that enable it to productively address a wide range of problems of social concern, including those traditionally considered to be the province of the social sciences, human services and information technology, as distinct from the decision sciences. We explore the relationship between Community OR and these domains in the remainder of this paper, which is organized as follows. Section 2 discusses challenges and opportunities for Community OR in research and practice. Section 3 presents emerging trends, primarily in the decision sciences and related fields, where OR is well positioned to have a substantial impact in the shorter term, and where Community OR might make a useful contribution. Section 4 discusses new frontiers: primarily areas distinct from the decision sciences, where there are longer-term prospects for Community OR's impact in practice and scholarship. Section 5 concludes with elements of a research agenda built upon the previous reflections.

# 2. Challenges and opportunities

Community operational research faces a number of barriers to widespread acceptance in teaching and research in decision science, and impact in practice commensurate with its social justice motivations. First, ordinary citizens may lack the expertise needed for data-driven problem solving, and therefore require considerable support (Gregory & Atkins, 2018; Ritchie, 2004). This is arguably one reason why there is more of an emphasis in Community OR on the participative use of qualitative methods. Also, Community OR often requires organized and sustained participation among multiple stakeholders for problem identification, formulation and solution (Gregory & Midgley, 2000; Taket & White, 2000), which may run counter to a tendency to rely on government and non-profit organizations to take the lead. Updating and managing socio-

technical systems is difficult, and is more commonly performed by trained practitioners working in established organizations. Knowledge generated by Community OR studies does not necessarily lead to the production of expert prescriptions in the sense traditionally understood within operational research, industrial engineering and related fields; rather, the practitioner may facilitate a process of learning that flows seamlessly into decision making without any need for expert recommendations (for examples, see many of the 26 case studies of practice in the book edited by Bryant, Ritchie, & Taket, 1994). Of course, most studies in the decision sciences with an application focus aspire to implementation as an end goal. However, special interests inside and outside the community may divert energy and enthusiasm towards aims not always shared by local actors. Stakeholder participation can be an antidote to this (Rosenhead & Mingers, 2001), but there also has to be a recognition that, in some contexts, there are stakeholders who try to get their way through manipulation or coercion rather than through engagement in free and fair dialogue (Jackson, 1991, 2006; Midgley, 1997), so this can make the design of Community OR projects and pathways to implementation quite complex. Online engagement is now relatively common in the current era of social media, especially with spatially dispersed communities, but there are barriers to the use of this in Community OR: it is not easy to translate the energy created online into in-person activism, especially if it requires sustained local action. Finally, there are few innovations within Community OR that have as high a profile within popular discourse as those associated with more traditional conceptions of OR, which reflect a traditional efficiency-enhancing approach and are rooted in the metaphors of logistics and business operations. Examples of these include manpower scheduling, revenue management, vehicle routing and production and operations management. Greater visibility is needed for our work.

However, despite the above challenges, Community OR and Community-Based Operations Research may also benefit from a number of opportunities to transform the decision sciences. Operational research for the public good, especially to benefit resourceconstrained and mission-driven nonprofit organizations, has received substantial visibility through a student paper contest sponsored by the Institute for Operations Research and the Management Sciences called 'Doing Good with Good OR' (INFORMS 2016a) as well as an edited volume sponsored by this society that is dedicated to public applications of operations research (Kaplan, 2015). There are also volunteer-driven initiatives of professional societies in the US ('Pro Bono Analytics', INFORMS 2016b) and the UK ('Pro Bono OR', Operational Research Society 2016). Scholars such as Mettler (2011) have demonstrated the importance of the 'submerged State' in diverting public benefits to most-privileged populations, providing a basis for Community OR researchers and practitioners to design interventions to reconcile technocratic/managerial understanding with people's real-world concerns. Current research in e-government and e-governance (Manoharan, 2015; Chen and Ahn, 2017) demonstrates the potential for internetenabled applications to provide high quality and rapid response services that can increase the level of trust between citizens, nonprofits and government. Initiatives and applications such as these may increase the likelihood that Community OR initiatives can be successful. These promising trends require an increased awareness by citizens of root causes of social concerns and the potential of localized direct action to address them, but the recent elections in the US and the EU ('Brexit') referendum in the UK may cast doubt on the willingness of many citizens to examine systemic barriers to an improved quality of life. In the face of this, we suggest that the primary opportunity for Community OR to increase its breadth and impact are specific application areas for which practitioners may provide novel and highly influential insights, strategies and operational recommendations. By doing so, we argue that Community

OR, and indeed OR more generally, may increasingly be seen as an attractive and easily-understood means by which to improve the quality of life of individuals and communities in ways not limited to market transactions and the activities of large organizations. We describe these application areas in the two sections that follow.

## 3. Emerging trends

Over the past 15–20 years, new areas within operational research have emerged to respond to contemporary issues in the public and private sectors that go beyond the traditional core foci of OR. As these new domains have become more closely associated with the mainstream of OR, there are increasing opportunities for Community OR to address them by adapting models, analytical methods and methodologies to strengthen a focus on community engagement and social change. Community OR practitioners can therefore make worthwhile contributions in the new areas while simultaneously moving our specialism more into the mainstream of OR. To us, the three most interesting emerging trends where Community OR can contribute something new, or gain something, are disaster planning, analytics and Behavioral OR. These are discussed below.

#### 3.1. Disaster planning

Recent work in disaster planning has critiqued common assumptions about expertise, and has focused on the role of communities in such planning. Auf der Heide (2006) reviews the literature and practice of medical planning for disasters. He shows that, while it is commonly assumed that trained emergency personnel carry out field operations, in most cases the initial rescues are done by survivors themselves (p. 36). Eisenman, Cordasco, Asch, Golden, and Glik (2007) attempt to understand the evacuation decisions of New Orleans residents in the context of Hurricane Katrina by looking at their interviewees' community ties. The researchers use a grounded theory approach to analyze responses from residents of Houston's major evacuation centers. In their discussion, they insist that evacuation must be studied much beyond the individual level, as "broad networks of families and friends create demands on participants" (Eisenman et al., 2007, p. 113).

Another promising avenue of research has been opened by Houston et al. (2015), who have developed a framework for using social media in disasters - with the main goal of implementing beneficial social media processes at all levels, including improving community resilience and reconnecting the community post-disaster. However, their section on how to heal fractured community links is rather short, especially in light of the recognized potential of social media uses (Houston et al., 2015, p. 15). There is room for much development here. In addition, Becker, Matson, Fischer, and Mastrandrea (2014) have examined stakeholder engagement and quantitative analysis for pre-disaster planning.

As a special case of disaster planning, humanitarian logistics applications tend to be conceived at the systems level and rely on mathematical modeling and solution algorithms associated with traditional OR (Duran, Ergun, Keskinocak, & Swann, 2012, Ekici, Keskinocak, & Swann, 2014; Liberatore, Ortuño, Tirado, Vitoriano, & Scaparra, 2014). One humanitarian logistics application, however, emphasizes appropriate information technology to support 'last mile' distribution of goods in communities with compromised infrastructures, developed in conjunction with local stakeholders (Ergun, Guyi, Heier-Stamm, Keskinocak, & Swann, 2014). While there has been a Community OR project on disaster planning (Gregory & Midgley, 2000), this predates the bulk of research in humanitarian logistics and does not speak to its focus on solving technical problems regarding the optimal positioning of goods

and services pre- and post-disaster. Community OR can enable researchers and practitioners to integrate stakeholders' experiences and preferences directly into an enhanced model development process that highlights difficulties in on-the-ground, disaster-related logistics. These difficulties include fear, uncertainty, corruption, political oppression and so on (Munday, 2015).

#### 3.2. Analytics

The discipline known as analytics comprises three distinct tasks: descriptive analytics, or the study of systems, organizations and phenomena according to historical data; predictive analytics, or the informed estimation of future values of variables or configurations of systems to aid in the anticipation of as yet unknown events; and prescriptive analytics, or the design of policies, guidelines or practices based on optimal or best possible values of decision variables assumed to be under the control of the modeler (Liberatore & Luo, 2010). Motivated by the explosion of data from processes and devices, the business process redesign movement and the widespread availability of sophisticated software, analytics has in many ways become the public face of the professions known heretofore as 'operational research' and 'management science'. Of course, the relabeling of disciplines and research communities is always contentious because professional identities are at stake, and there is clearly an ambiguous relationship between OR and analytics (Mortenson, Doherty, & Robinson, 2014). Popular treatments of analytics are numerous, including Nussbaumer Knaflic (2015) and Siegel (2016). The Institute for Operations Research and the Management Sciences, for example, has developed an analytics credentialing program for practitioners (INFORMS 2016c) and a new analytics maturity model to facilitate organizational redesign through analytics (INFORMS 2016d).

Though analytics is most often conceived as a quantitative domain, recent work has emphasized the role that problem structuring methods (PSMs) and other qualitative methods may play in it (Ranyard, Fildes, & Hu, 2015). Community OR, which makes frequent use of PSMs, may benefit from emphasizing its relationship with analytics, interpreted broadly as ways to solve practical problems using diverse methods, some of which involve quantitative data. Indeed, an association between the two areas has already been made by Hindle and Vidgen (2018). Community OR may make contributions to analytics by emphasizing questions of

- What data can and should do for individuals, communities and organizations;
- Whether concepts such as effectiveness, social impact, institutional challenges and the like are, or can be, taken seriously by decision makers seeking to quantify various aspects of an analysis; and
- Most importantly for Community OR, how community residents themselves (and their representatives) can work with organizations to define, collect and analyze data that are relevant to their own lives (see the subsection on 'big and difficult data' below).

Some of these questions have been addressed in the context of volunteer consulting engagements sponsored by Pro Bono Analytics (INFORMS 2016b) and Pro Bono OR (ORS 2016).

## 3.3. Behavioral OR

While humanitarian logistics and analytics are high profile new areas where Community OR might make substantial contributions, the third emerging trend to be discussed, Behavioral OR (e.g. Franco & Hämäläinen, 2016; Hämäläinen, Luoma, & Saarinen, 2013), is one where the benefit is most likely to be in the other direction: we argue that Community OR, along with OR more

broadly, can learn something significant from Behavioral OR. Indeed, this learning has already started (Velez-Castiblanco, Brocklesby, & Midgley, 2016).

There are actually several different strands of Behavioral OR (Becker, 2016; Franco & Hämäläinen, 2016), but the one we are most concerned with here involves the close study of participant and practitioner interactions and communications to identify what, in the OR modeling process, makes a critical difference in terms of participant learning and decision making (e.g. Brocklesby, 2016; Luoma, 2016; Scott, Cavana, & Cameron, 2016; Tavella & Franco, 2015; Thompson et al., 2016; White, Burger, & Yearworth, 2016). By recording and viewing the micro-level interactions in OR workshops, behavioral researchers aim to discover critical factors for success and failure so that future OR practice can be designed to account for these.

As far as we are aware, there has been only one application of behavioral analysis to a project explicitly identified as Community OR: see Foote, Ahuriri-Driscoll, Hepi, Midgley, and Earl-Goulet (2016) for details of the project, and Velez-Castiblanco et al. (2016) for the analysis. Velez-Castiblanco et al. (2016) examined how the OR team collaborated on the design of a mixed-methods intervention, and were able to show that the design process was very different from the rather sanitized accounts of method selection usually presented in the OR literature (also see Keys & Midgley, 2002, Midgley, 2000, and Ormerod, 2014, for critiques of this sanitization). The design process involved the deployment of a great deal of tacit knowledge as well as various tactics of influence and persuasion. The authors' theory of 'boundary games' provides a way of thinking about communications that

"...can support greater mindfulness, both when OR teams are designing an intervention and when the intervention is being undertaken. By 'mindfulness', we mean conscious reflection in the context of the flow of dialogue and action, which can augment the tacit knowledge that is inevitably a major feature of OR processes" (Velez-Castiblanco et al., 2016, p.979).

Future analyses of Community OR projects could feature examinations of the interactions of participants around models; these could be significant in the evaluation and accelerated improvement of practice.

## 4. New frontiers

In contrast with the three domains described above, there are a number of other research areas which have had identities substantially or largely distinct from OR, but which have characteristics that are supportive of innovations within Community OR (and CBOR). Community OR draws its power from the insights it may provide on contemporary policies, application contexts and technologies that have an impact on spatially distinct and constrained groups of people and infrastructures. Here we want to discuss how Community OR may be applied to a number of new frontiers, and what the discipline may learn from these new applications. Clearly, there are many more new frontiers than can be covered by a single paper, but we have selected nine that seem to us to be particularly promising, either because work is already going on to inform them with Community OR (e.g., community-based intervention in developing countries), or because people in those areas share common values with Community OR practitioners (e.g., working in the service of indigenous people on issues of concern in their communi-

We are not aware of any single theoretical framework that we could have used to select the frontiers for review, as there are diverse social forces shaping them and no one theory neatly covers them all. Some of these social forces include urbanization and the mitigation of slums, especially in developing coun-

tries (e.g. Davis, 2007); access to education and education reform (e.g. Adamson, Astrand, & Darling-Hammond, 2016); community health, including food security and access to health care (e.g. Galea and Vlahov, 2005); shrinking cities and municipal decline, especially in developed countries (e.g. Weaver, Bagchi-Sen, Knight, & Frazier, 2017); neighborhood resilience and disaster planning (e.g. Hicks Masterman et al., 2014); crime, disorder and community safety (e.g. Bowden, 2014); sustainable cities (McLaren & Agyeman, 2015); diversity, inclusion and multiculturalism (e.g. Vertovec and Wessendorf, 2010) and many others. Indeed, if we had artificially imposed a single framework on our analysis, it could have resulted in the omission of some important frontiers and an over-emphasis on others. We therefore rely primarily on our knowledge and experience for our selection.

#### 4.1. Urban planning and community development

Urban planning, community development, urban affairs and related fields are focused on developing strategies to make homes, neighborhoods and cities better places to live for as many people as possible. This can be done by permitting, encouraging or forbidding certain types of physical infrastructure (urban planning); helping local residents advocate for their needs; developing local services and institutions (community development); and providing guidance and insight regarding all manner of products and services intended to meet the needs of urbanized communities (urban affairs) (see, e.g., Hall & Tewdwr-Jones, 2010; Levy, 2017). Since these fields have the improvement of life through peoples' daily activities as a core concern, there would seem to be a role for Community OR. Indeed, Johnson and colleagues have adapted principles of Community OR to address community responses to the housing foreclosure crisis, both at the higher level of project design (collaborating with community partners to identify and set research agendas; see Turcotte, Johnson, Chaves, Drew, & Sullivan, 2015) and at the lower level of executing particular research designs. In terms of the latter, examples are identifying local values associated with community revitalization and foreclosure response (Keisler, Turcotte, Drew, & Johnson, 2014) and developing a novel metric for community development that links strategy and impact (Johnson, Drew, Keisler, & Turcotte, 2012). Johnson, Hollander, and Davenport Whiteman (2015) have also employed some principles of Community OR to design and evaluate decision models for nontraditional local development to counter blight, vacancy and abandoned properties.

However, despite a wide range of potential application areas within the service sector (including transportation and warehousing, information and communication, human health and social assistance, financial and insurance services, and many more), documented applications of non-Community OR in the service sector appear to be overwhelmingly concentrated on traditional quantitative, mathematical model-driven approaches (Zhong, Karner, Kuby, & Golub, 2017; Xing, Li, Bi, Wilamowska-Korsak, & Zhang, 2013). Community OR principles are actually quite prominent in contemporary treatments of community development (see e.g. Defilippis & Saegert, 2012) and new initiatives to build collaborations between researchers and practitioners for community development and social change exist (URBAN 2016), although they are not usually named as 'Community OR' in teaching and research contexts. Community OR may benefit greatly by emphasizing connections with urban/city planning and community development, and connecting well-understood methods for community engagement and design with decision science principles of problem identification, formulation and solution that have an emphasis on process learning and stakeholder impact rather than mostly on technical issues of mathematical modeling and algorithm design.

#### 4.2. Information systems and information technologies (IS/IT)

IS/IT is a well-studied domain, both in stand-alone academic disciplines and university departments and colleges, and is an active area of inquiry within the decision sciences (see e.g. Information Systems Research, http://pubsonline.informs.org/journal/ isre). However, there has been relatively little attention paid by these disciplines to the design of community-engaged methods for problem solving in the OR tradition, and conversely relatively little attention has been paid within Community OR to IS/IT as either vehicles for solution implementation or domains within which intensive community engagement might be performed (Yearworth and White, 2018). There are, of course, exceptions: for instance, Córdoba and Midgley (2003, 2006, 2008) show how IS planning can put stakeholders (including those in the community) front and center through an approach based on critical systems thinking; and Barros-Castro, Midgley, and Pinzón (2015) apply a similar approach to the engagement of school children and teachers in the design of computer-supported collaborative learning programs. With this in mind, perhaps one of the most fruitful areas of potential interchange between Community OR and IS/IT researchers is that of citizen engagement. For both fields, acquiring knowledge on the ultimate 'end users' and ensuring their empowerment are central concerns (e.g. Córdoba-Pachón, 2010).

Another illustrative example comes from Barrett, Oborn, and Orlikowski (2016), who examine how value is created in online communities over time. They argue that researchers need to move away from considering online participants as largely homogenous, and towards identifying stakeholder groups and key participants. They do precisely that in a case study of an online healthcare community group, and such work showcases the potential of IS research for community engagement. Lopez (2015) does something related but different, while looking at the online behaviors of urban communities targeted by participatory information systems programs. She finds that the geographic targeting scope matters a great deal (local versus hyper-local), but also that off-site communication is essential to IS development, and the greatest challenge to online community sustainability is residential instability. To avoid marginalization, designers of participatory information systems must engage transient populations as well as those remaining for longer periods of time. In any case, research indicates that information systems seeking to engage small, urban communities need to be designed differently from sites with a global reach. Raymaker (2016) is an example of a practitioner who explores the latest IS research on direct engagement using critical systems thinking and community-based participatory research. Her study is an exploration of the development process of a healthcarefocused web site for autistic end users, but the implications for further research are immediately obvious: what would be different if the engagement was directed at other populations or organizational contexts?

Thus, Community OR could transform IS/IT by emphasizing community-engaged methods for systems design and implementation, drawing connections between technical innovations and conventional notions of usability, and the expressed needs of especially disadvantaged end-users whose low income, lack of social status and influence may make them more often seen as consumers of IS/IT innovations rather than sources of such.

## 4.3. Big and difficult data

The past decade or so has seen an explosion of research in the area of 'big data', commonly understood to be the collection of very large datasets routinely generated through information systems such as point-of-sale systems, social media, public surveillance and the 'internet of things' (Bollier, 2010). Big data can be

seen as an aspect of analytics, but the focus here is more on the sources of the data, and the special problems associated with handling huge repositories of data, constantly refreshed from diverse sources in real time. While datasets and analyses of big data are usually dominated by experts working at a distance from local communities, there is growing evidence of community participation in large scale data analysis, through crowd-sourcing and community activism (Calvard, 2016). Moreover, researchers are increasingly exploring issues related to the curation of large datasets of public and local interest (Bertot, Butler, & Travis, 2014) and community collaborations to extract alternative meanings from large datasets (Couldry & Powell, 2014).

Another approach to data analytics and community engagement arises from the notion that, in many cases, the data that are most relevant to community needs, especially disadvantaged and lower-income communities, are not 'big' at all, but challenging because of a lack of consensus on what data elements should be collected, from what sources and put towards what ends (which explains why this section has a title distinct from simply 'big data'). Also, even modestly-sized datasets can tax the capacity of missiondriven nonprofit organizations (Boland, 2012; Johnson, 2015; Johnson et al., 2015). The research on this conveys in aggregate a substantial opportunity for Community OR to fully engage in research on data science and analytics, using our discipline's unique perspective on local agency and a critical approach to identify novel applications for data collection, analysis and use for local development. One contemporary application of community data analytics, for example, is the issue of defining specific metrics for measuring the impacts of local economic development, with data collected by grass-roots organizations that are distinct from those mandated by local government (Johnson & Jani, 2016).

#### 4.4. Smart cities

Connecting with big data is the movement to harness large datasets to improve the operations and management of government and services within neighborhoods and cities through advanced technologies. A 'smart city' is defined as "a synthesis of hard infrastructure (or physical capital) with the availability and quality of knowledge communication and social infrastructure. The latter form of capital is decisive for urban competitiveness" (Batty et al., 2012, p. 486, citing Caragliu, Del Bo, & Nijkamp, 2009). Most smart city applications, such as the real-time analysis of mass transit data for better prediction of ridership and congestion (see e.g. Batty, 2013), appear to be conventional applications of centralized, expert-driven analyses. However, Batty et al. (2012) acknowledge the potential for democratic participation via 'citizen science' regarding the nature, content and use of large datasets for urban operations and management (see Gregory & Atkins, 2018, for some reflections on the potential for connecting Community OR with cit-

Since smart cities have the potential to affect the lives of their residents at all times and in all places, there appear to be significant opportunities for Community OR to enable diverse stakeholders to influence the ways in which smart cities are designed and implemented, and to apply stakeholder engagement to define the real-life problems they purport to solve. In particular, Community OR can challenge common notions of technology as a mostly unalloyed good, emphasizing the role that smart city-focused technologies can play in expanding the reach of the surveillance state and highlighting class and social disparities in access to and use of smart city technologies and data (see, e.g. White & Trump, 2016).

IBM is a company that has invested significantly in the area of smart cities (e.g. Dirks & Keeling, 2009), and one of the authors (Gerald Midgley) was engaged with them for several years on the sticking points for implementation, which often concern the gover-

nance of initiatives and not the technology. Here, the emphasis of Community OR on community and stakeholder engagement could make a significant difference to both the design of these initiatives to meet community concerns, and their eventual implementation. Since much of the literature on both Community OR and smart cities has been influenced by systems thinking, there is already enough of a common language to make a start.

#### 4.5. Resilient cities

The notion of resilient cities addresses a myriad of contemporary challenges (such as economic development, social polarization and segregation, as well as climate change and ecological degradation) through the notion of planning, adaptation and response to immediate and long-term threats to human and community health (Spaans & Waterhout, 2017). Resilient cities are well-positioned to direct intervention in physical and social infrastructure in urbanized areas to redress inequalities and structural flaws. The importance of resilient cities is represented by the worldwide '100 Resilient Cities Program' sponsored by the Rockefeller Foundation. Research programs in resilient cities abound, including Boston in the USA (Martin, 2015); Rotterdam in the Netherlands (Spaans & Waterhout, 2017); Dhaka in Bangladesh (Walters, 2015); and Brisbane in Australia (Walters, 2015). These studies take as given the importance of community engagement and participation in risk assessment and planning for, and adaptation to, long-term changes in climate and the economy that could reduce quality of life, as well as displace and put at risk the lives of millions of people. They directly engage the notions of stress and trauma, not just as the result of discrete disasters, but as indicators of reduced well-being in response to chronic mal-adaptations of urban areas to global change, experienced locally. There appears to be a significant opportunity for Community OR to articulate community values, structure objectives and develop interventions in close cooperation with affected communities, especially those at greatest risk of harm due to low income, lack of political influence, low levels of social engagement, etc. See Helfgott (2018) for a project taking a significant step forward in this direction.

# 4.6. Developing countries

Countries in the 'global South' face a much different set of political, social and economic challenges than the developed countries in which OR was started (Rosenhead, 1993). From the increased severity of climate-change-related extreme weather events, to high levels of internal and external human displacement due to war, political instability and food insecurity, to daily life challenges arising from poverty, disease and a lack of good governance, developing country issues might initially seem to be too large and systemic to be addressed in a substantive way by Community OR, as opposed to well-established disciplines such as international development, human security and global governance. However, there has been consistent progress towards the creation of OR infrastructures in education and research in developing countries, as exemplified by Caulkins, Eeelman, Ratnatunga, and Schaarsmith (2008) and Maposa, Cochran, and Lesaoana (2016). Caulkins et al provide specific examples of OR teaching that can be easily adapted to resource-constrained environments, and which are responsive to decision problems arising from the African experience. Maposa et al. (2016) present a more traditional quantitative modeling-based approach to extreme weather event forecasting and response.

The literature on Community OR engagements in developing countries includes work by Ochoa-Arias (1994, 2004) in Venezuela; White (1994) in Belize; Sova, Helfgott, Chaudhury, Matthews,

Thornton, and Vermeulen (2015) and Helfgott (2018) in Nepal; Tirivanhu, Matondi, and Sun (2016) in Zimbabwe; Espinosa and Duque (2018) and Pinzón-Salcedo and Torres-Cuello (2018) in Colombia; Mwiti and Goulding (2018) in Kenya; Romm (2018) in South Africa; Gomes, Hermans, and Thissen (2018) in Bangladesh; Burns (2018) in Myanmar; and Ufua et al. (2018) in Nigeria. There is also a more general review of OR applications in development (White, Smith, & Currie, 2011); a set of ideas for adapting Community OR to the needs of developing countries (Rosenhead, 1993); and a proposal for a new conception of public health in development based on Community OR principles and practice (Thunhurst, 2013). These works emphasize the roles that community engagement, qualitative methods, problem structuring and values-explicit inquiry may play in developing country settings.

# 4.7. Diversity and inclusion

Diversity can be broadly understood as encompassing individual characteristics that are often viewed or treated as markers of social difference or collective identity, as well as internal individual characteristics that may reflect personal understandings of the world. The latter are often referred to as cognitive diversity. Diversity always exists in social systems. Inclusion, on the other hand, often needs to be consciously enabled. In order to leverage diversity, an environment must be created where people feel safe, supported, listened to, valued and able to do their personal best. This is often a 'wicked problem' (to borrow a concept from Rittel & Webber, 1973), in the sense that inclusion initiatives tend to have numerous stakeholders, may interact with a variety of problems and attempted solutions, and trade-offs between values are difficult to state (or when stated, may be difficult for some stakeholders to accept).

There are many diversity and inclusion problems amenable to solution using traditional OR, such as public school assignment (Shi, 2015); college admissions (Chen & Kesten, 2016); and job interview process design (Johnson, Hekman, & Chan, 2016). However, the most challenging problems, particularly those requiring more comprehensive attempts to solve failures of inclusion, may not be accommodated by traditional approaches, and here we refer to things like gender diversity quotas, job guarantee programs and universal basic income design, for instance. It is difficult to imagine any of these being tackled competently without meaningful community engagement and without a critical awareness of the strengths and weaknesses of different approaches.

Examples of applications of Community OR to diversity and inclusion are presented by Pindar (1994), who focuses on racial harassment; Cohen and Midgley (1994) and Midgley and Milne (1995), who look at the marginalization and inclusion of people with mental health problems; Gregory, Romm, and Walsh (1994) and Gregory and Romm (2001), who discuss the empowerment of blind and partially sighted health service users; and Boyd, Brown, and Midgley (2004), who explain the design of a Community OR process that put the perspectives of homeless children at the heart of the development of services to meet their needs. This is an area that has already been of significant concern in Community OR; for a wider set of readings, see various chapters in Johnson (2012a), Midgley and Ochoa-Arias (2004b) and Ritchie et al. (1994). However, there is still the potential for further innovation, especially to build a practice-relevant theory of inclusion as a generic issue.

COR has the potential to build interventions around *critical* perspectives on diversity and inclusion. These interventions can account for many issues, such as power relationships that characterize the institution or phenomenon of interest; conflicts between stakeholders with different perspectives; the choice of preserving or replacing the current organizational structure in the interests

of social justice; how community engagement should be defined; and the existence of roles for systems thinking and problem structuring methods in developing novel solutions to issues of diversity and inclusion (Johnson, 2016). The recent unveiling of a comprehensive and radical policy platform for the US-based Black Lives Matter movement (The Movement for Black Lives, 2016) provides a promising opportunity for Community OR practitioners to engage with local activists to develop interventions that support efforts for social change. One aspect of Community OR that could be particularly useful in this area is the theory of boundaries and marginalization processes that has informed a number of interventions (e.g. Barros-Castro et al., 2015; Boyd et al., 2004, 2007; Córdoba and Midgley, 2003, 2006, 2008; Foote et al., 2007; Midgley et al., 1998, 2007; Midgley, 2000, 2006, 2015; Midgley & Shen, 2007; Midgley & Pinzón, 2013; Shen & Midgley, 2015; Ufua et al., 2018).

#### 4.8. Environmental issues

Traditionally concerned with stewardship and sustainability, environmental policy and action has broadened its focus to address systemic issues related to climate change, urban resilience and human adaptation. Central to all of these application areas is the role of community and stakeholder participation in both agenda setting and individual/collective action. Ulrich (1993) discusses the need for systems thinkers to engage the ecological movement without false pretensions that systems/OR can provide comprehensive analyses; rather, we should be aware of our boundary judgments and the values that inform them. Also see Midgley (1994) for a discussion of the frequent marginalization of environmental issues due to overly narrow boundaries defining economic and social concerns. Midgley and Reynolds (2001, 2004a,b) present an agenda for change in OR to meet the needs of environmental management, which includes a greater focus on stakeholder and community engagement. Waltner-Toews, Kay, Murray, and Neudoerffer (2004) offer a new Community OR methodology to integrate community engagement with scientific analysis in projects where both environmental and social values need to be accounted for in development proposals.

More recent OR on environmental issues includes the work of Schafer and Gallemore (2016) on the use of multi-criteria decision analysis for agenda setting in natural resources project funding; Pimentel, Santibañez Gonzalez, and Barbosa (2016) addressing modeling principles for decision support system development, focusing on environmentally friendly mining; and adaptations of principles of problem structuring methods and value-focused thinking for Life Cycle Sustainability Assessment in waste management systems (Souza, Rosenhead, Salhofer, Valle, & Lins, 2014). In this body of research, however, the role of community members, as opposed to planners and managers, does not appear to be prominent, so there is an opportunity here for Community OR practitioners to highlight the benefits of community-based intervention design, implementation and evaluation.

#### 4.9. Indigenous people's issues

While much of the literature on Community OR originates from the UK and US, as the movement spreads into other countries with different cultural histories, Community OR theory and practice will come into contact with indigenous people. Indeed, there is already a small but growing literature on OR practitioners working with indigenous communities (Ahuriri-Driscoll, Baker, & Midgley, 2005; Ahuriri-Driscoll & Foote, 2016; Brocklesby & Beall, 2018; Espinosa & Duque, 2018; Foote et al., 2005; Foote, Hepi, Rogers-Koroheke, & Taimona, 2017; Hepi, Foote, Rogers-Koroheke, & Taimona, 2007; Jellie et al., 2003; Midgley et al., 2007; Morgan and Fa'aui, 2018).

The term 'indigenous' refers to the 'first people' in any given country, who were there before colonizing forces arrived (Smith, 1999). There are various people around the world who are trying to preserve their native cultures despite sometimes overwhelming pressure to give up their ancestral lands, languages and identities – and there are others (such as Māori in New Zealand) who have survived this cultural imperialism and are now resurgent, despite continuing inequality and disadvantage. They are developing their languages and identities in new directions while still firmly anchored in their cultural history and traditions.

Working with indigenous people raises issues that are not encountered in any other setting. For a start, Western science, including the language and methodologies of OR, can be viewed as an instrument of domination because, historically, indigenous people have been subject to the 'objective' gaze of researchers who viewed their cultures as alien curiosities or ridden with primitive superstitions (Smith, 1999). This attitude has left deep suspicions amongst indigenous communities that scientists and operational researchers are at best going to take from them (in the form of publications and reputation) without giving anything in return, and at worst are going to assume that they have a superior rationality that validates the imposition of their own problem definitions and solutions without proper community engagement. These suspicions are amplified when the OR practitioners are employed by institutions that were originally founded by colonists (such as government departments), and non-indigenous Community OR practitioners therefore have to be strongly aware of identity issues and build relationships with communities over time and with cultural sensitivity (Midgley et al., 2007; Walsh, Kittler, & Mahal, 2018). Indeed, in recent years, there have been movements in some indigenous communities to create their own methodologies, grounded in their own cultures. A good example is Kaupapa Māori in New Zealand (Bishop, 1996; Smith, 1999), which involves research by Māori, for Māori. Non-Māori OR practitioners can be involved, as long as the leadership sits with Māori themselves. This poses a significant challenge to OR practitioners, who need to negotiate everything, including whether they will be allowed to publish under the banner of Community OR.

Because of this challenge, the learning opportunities for Community OR practitioners are substantial (Brocklesby and Beall, 2018). First of all, when working in communities where there is a strong culture of indigenous research, lessons can be learned about the full potential of community leadership in co-creating OR projects (also see Ackoff, 1970, who worked in a non-indigenous context, but one where the project was constructed to give local people full leadership responsibilities). The potential is there for much more exciting and community-relevant outcomes, and for alternative theories, methodologies and practices of Community OR that are culture-specific. Also see various discussions of culture-specific Systems/OR methodologies in non-indigenous, non-Western contexts (e.g. Gu & Zhu, 2000; Li & Zhu, 2014; Midgley and Wilby, 1995, 2000; Midgley, Gu, & Campbell, 2000; Midgley & Shen, 2007; Murthy, 1994; Shen & Midgley, 2007a, 2007b, 2015; Tan, Watson, & Wei, 1995; Wang, 2000; Zhu, 2000).

The other significant opportunity for learning is to experience indigenous methodologies and methods and see whether they are transferable or adaptable to other cultures elsewhere in the world—with the permission of their indigenous curators, of course, and giving credit to the original contexts in which these approaches were developed. This could substantially enrich our Community OR toolkits, and is ethical as long as the sharing is voluntary and two-way; i.e. it is a case of indigenous communities enriching their own practice on their own terms as well as non-indigenous communities learning from them.

**Table 1**Summary of emerging trends and new frontiers in Community OR.

Emerging Trends	
Disaster planning	Community OR may enable an integration of various stakeholders' experiences and preferences, especially neighborhood-level community preparation, directly into an enhanced model of disaster planning and response.
Analytics	Community OR emphasizes questions of what data can and should do for individuals and communities, and how community residents themselves can work to define, collect and analyze data that are relevant to their own lives.
Behavioral operational research	Community OR can be enhanced by Behavioral OR analyses that demonstrate what works best in participative modeling processes, and how meaningful engagement can be enhanced.
New Frontiers	
Urban planning and community development	Many urban planning applications are still entirely quantitative, modeling-driven approaches; Community OR can identify local values associated with community revitalization and develop relevant metrics.
Information systems and information technology	Community OR can contribute to the IS/IT literature on citizen engagement by emphasizing community-engaged methods for ICT planning, design and implementation.
Big and difficult data	Community OR has a unique perspective on local agency, and uses a critical approach to identify applications for data collection, analysis and use for local development. Community OR methods may be useful when there is a lack of consensus on sources, variables and uses for data.
Smart cities	Community OR may challenge notions of technology as a mostly unalloyed good, emphasize the role that smart city-focused technologies can play in expanding the reach of the surveillance state and highlight class and social disparities. It may also provide useful methods for citizen and stakeholder engagement to improve the governance of smart city initiatives and make them more responsive to grass-roots community concerns.
Resilient cities	Community OR may help structure objectives and develop interventions in close cooperation with affected communities, especially those experiencing stress and trauma.
Developing countries	Community OR methods could be especially adapted to resource-constrained environments and to decision problems arising from the experiences of people in developing countries.
Diversity and inclusion	Issues of inclusion are often "wicked" problems requiring the involvement of numerous stakeholders who need to get to grips with multiple interactions with other problems as well as clarify and discuss conflicting and poorly-articulated values. Community OR can help define crucial notions of community engagement, and bring in useful systems theories of boundaries and marginalization processes.
Environmental issues	Community OR may highlight the special nature of community-based intervention design, implementation and evaluation; Community OR methodologies should be able to integrate community engagement with scientific analysis.
Indigenous people's issues	Community OR is concerned with meaningful engagement, co-creating interventions with community-based partners and using methodologies and methods for empowerment. All of these are essential to working with integrity in indigenous communities. In addition, Community OR researchers may experience indigenous methodologies and methods and examine (with the permission of their curators) whether they are transferable or adaptable to other cultures elsewhere in the world.

## 5. Conclusion

We have identified areas in which Community OR can enrich, and be enriched by, current research in multiple domains within and outside the decision sciences: disaster planning (including humanitarian logistics); analytics; behavioral OR; urban planning and community development; information systems and information technology; big and difficult data; smart cities; resilient cities; developing countries; diversity and inclusion; environmental issues; and indigenous people's issues. Across these domains, within and beyond OR, we have argued that all of the following aspects of Community OR can enrich our understanding of theory, methods and outcomes: its notions of intervention, local engagement and impact; its frequent concern for disadvantaged, underrepresented and underserved populations; its emphasis on problem solving processes as well as outcomes; its critical attitude and concern for ethics; its leaning toward qualitative and mixed method approaches; and, in general, its concern for community empowerment and social change. We have summarized these findings in Table 1

It seems to us that wider engagement (beyond clients) and a critical perspective are particularly important when developing new approaches to analytical thinking for creative problem solving across disciplines and applications. While these notions are

well accepted in some social science and transdisciplinary research communities, they are less commonly understood in the decision sciences – but this is where they arguably matter most, because the decision sciences are so concerned with application and impact. Thus, like Jackson (1987, 1988) and Midgley et al. (2018), we claim that a greater appreciation for the potential of Community OR principles, theory, methodology and methods – especially in relation to engagement and critical thinking – can enrich the decision sciences.

The analysis in this paper leads us to propose some potential implications for the theory, methods and practices within the decision sciences. First, the conception of 'community' can be broadened to address online as well as in-person communities (also see Yearworth and White, 2018); ones that are geographically concentrated as well as spatially dispersed; those that are defined by immutable versus changeable characteristics; and those that cohere around visible versus invisible characteristics. This issue is especially salient to urban planning and community development, information systems and information technology, diversity/inclusion and environmental issues. In these domains, questions of which stakeholder groups 'count', how they are affected by decision problems, and how their views can be incorporated into interventions are important and challenging.

Second, the notion of 'problem solving' can be broadened to encompass novel understandings of individual and collective values, motivations for action, cultural perceptions in organizations and systems, as well as more traditional notions of policies, prescriptions and new procedures. This notion is important in humanitarian logistics, for example, where questions about how problems should be solved, and what the impacts of a problem solution might be on affected groups (beyond technical and logistical concerns), could receive more attention than they do at present, especially in the US OR context. This notion is important in a different way for indigenous people's issues, where cultural conflict, and the contrast between the perspectives of the researcher and the researched, make traditional applications of OR problematic.

Last, the notion of a disciplinary 'home' or frame for inquiry can be loosened somewhat (especially in the US context) so that, for example, urban planning, analytics or information systems can become more accepting of Community OR thinking.

Regarding methods, we have argued that Community OR practitioners generally show a greater willingness than many others in the decision sciences to use mixed method designs to solve challenging problems; to connect interventions with systemic analyses wherever possible, rather than deal with superficial symptoms of deeper social problems; and to identify high-impact human outcomes, as compared with technical system change. We see these concerns as especially important for diversity and inclusion, environmental issues and working in developing countries; in these areas, multiple analytical methods are commonly applied within specific disciplinary domains, but less often imported across disciplinary boundaries. To take just one example, how could a traditional approach to diversity and inclusion in an organization, as enhanced through decision science principles, deal with the concerns of members of underrepresented or marginalized groups who may not actually participate within that organization at present? We suggest that any credible project addressing this question would have to engage communities in the manner that is common in Community OR projects.

Regarding practice, we are reminded of the importance of interrogating common, disempowering assumptions about the roles that communities and their representatives should play in relation to problem solving in the agencies that serve those communities. More openness and flexibility is needed, especially when identifying problems, to make sure that agencies are not missing crucial issues. Within projects, insights, modeling and suggestions for change need to be better connected to stakeholder values, to improve local relevance. Again, especially in the US context, we suggest that technological—managerial solutions are essential but not sufficient. These insights seem especially important to the analytics, smart cities and big data movements, which often appear more interested in technologies, markets and data than in ways that they can engage underrepresented communities and define outcomes in terms that are relevant to diverse groups of citizens.

These implications for theory, methods and practice may provide the basis for a research agenda that *engages Community OR* with the decision sciences more generally. Such an agenda should embrace concerns with inter- and trans-disciplinary inquiry, systems thinking, community engagement, equity and social justice, and the implementation of solutions that embrace changes to human as well as technological systems. The current fraught political environments in the US and the UK increase the importance of such values, and suggest that researchers outside the US and UK have a special opportunity to develop extensions to Community OR and the decision sciences to support local development and community empowerment based on empiricism and critical inquiry for improved problem solving.

#### Acknowledgment

Gerald Midgley's participation in writing this paper was partfunded by the UK Natural Environment Research Council (NERC) and the Economic and Social Research Council (ESRC) through project NE/L014211/1 under the Resource Recovery from Waste theme.

#### References

- Ackermann, F., Bawden, R., Bosch, O., Brocklesby, J., Bryant, J., Buede, D., et al. (2009). The case for soft O.R. OR/MS Today, 36(2), 20–21.
- Ackoff, R. L. (1970). A black ghetto's research on a university. *Operations Research*, 18, 761–771. doi:10.1287/opre.18.5.761.
- Adamson, F., Astrand, B., & Darling-Hammond, L. (2016). Global education reform:

  How privatization and public investment influence education outcomes. New York:
  Routledge.
- Ahuriri-Driscoll, A., Baker, V., & Midgley, G. (2005). Whānau and whānau development in te awakairangi: Te rūnanganui o taranaki whānui kit e upoko o te ika a māui inc. Whānau development and learning communities initiatives. Client Report for Te Puni Kökiri. Christchurch: Institute for Environmental Science and Research
- Ahuriri-Driscoll, A., & Foote, J. (2016). Windows, wheels and wai: Public policy, environmental health action and Māori community development- Implications for (eco)social work. In J. Maidment, & E. Beddoe (Eds.), New Zealand social policy for social work and human services: Diverse perspectives. Christchurch: Canterbury University Press.
- Alrøe, H. F. (2000). Science as systems learning: Some reflections on the cognitive and communicational aspects of science. Cybernetics and Human Knowing, 7, 57–78.
- Auf der Heide, E. (2006). The importance of evidence-based disaster planning. *Annals of Emergency Medicine*, 47(1), 34–49. doi:10.1016/j.annemergmed.2005.05.
- Bardach, E., & Patashnik, E. M. (2016). A practical guide for policy analysis: The eightfold path to more effective problem solving (5th Edition). Los Angeles: CQ Press.
- Barrett, M., Oborn, E., & Orlikowski, W. (2016). Creating value in online communities: The sociomaterial configuring of strategy, platform, and stakeholder engagement. *Information Systems Research*, 27(4), 704–723. doi:10.1287/isre.2016. 0648.
- Barros-Castro, R., Midgley, G., & Pinzón, L. (2015). Systemic intervention for computer-supported collaborative learning. Systems Research and Behavioral Science, 32(1), 86–105. doi:10.1002/sres.2220.
- Batty, M. (2013). Big data, smart cities and city planning. Dialogues in Human Geography, 3(3), 274–279. doi:10.1177/2043820613513390.
- Batty, M., Axhausen, K. W., Giannotti, F., Pozdnoukhov, A., Bazzani, A., Wachowicz, M., et al. (2012). Smart cities of the future. The European Physical Journal Special Topics, 214(1), 481–518. doi:10.1140/epjst/e2012-01703-3.
- Becker, A., Matson, P., Fischer, M., & Mastrandrea, M. (2014). Towards seaport resilience for climate change adaptation: Stakeholder perceptions of hurricane impacts in Gulfport (MS) and Providence (RI). Progress in Planning, 99, 1–49. doi:10.1016/j.progress.2013.11.002.
- Becker, K. H. (2016). An outlook on behavioural OR Three tasks, three pitfalls, one definition. European Journal of Operational Research, 249, 806–815. doi:10.1016/j. ejor.2015.09.055.
- Bertot, J. C., Butler, B. S., & Travis, D. M. (2014). Local big data. In J. Puron-Cid, S. Robinson, J. Zhang, & J. R. Gil-Garcia (Eds.), Proceedings of the 15th annual international conference on digital government research '14 (pp. 17–23). ACM. doi:10.1145/2612733.2612762.
- Bishop, R. (1996). Collaborative research stories: Whakawhanaungatanga. Palmerston North: Dunmore Press.
- Boland, S. (2012). "Big data for little nonprofits". Nonprofit Quarterly, November 28, 2012. Web: http://nonprofitquarterly.org/management/21410-big-data-for-little-nonpr. Retrieved August 18, 2016.
- Bollier, D. (2010). The promise and peril of big data. Washington, DC: The Aspen Institute https://assets.aspeninstitute.org/content/uploads/files/content/docs/pubs/The\_Promise\_and\_Peril\_of\_Big\_Data.pdf [accessed 18 August 2016].
- Bowden, M. (2014). Crime, disorder and symbolic violence: Governing the urban periphery. Houndmills, UK: Palgrave Macmillan.
- Boyd, A., Geerling, T., Gregory, W., Kagan, C., Midgley, G., Murray, P., et al. (2007). Systemic evaluation: A participative, multi-method approach. *Journal of the Operational Research Society*, 58, 1306–1320. doi:10.1057/palgrave.jors.2602281.
- Boyd, A., Brown, M., & Midgley, G. (2004). Systemic intervention for community OR: Developing services with young people (under 16) living on the streets. In G. Midgley, & A. E. Ochoa-Arias (Eds.), Community operational research: OR and systems thinking for community development. New York: Kluwer. doi:10.1007/ 978-1-4419-8911-6 10.
- Brocklesby, J., & Beall, E. (2018). Processes of engagement and methodology design in community operational research: Insights from the indigenous peoples sector. European Journal of Operational Research in press. doi:10.1016/j.ejor.2017.07. 053.
- Brocklesby, J. (2016). The what, the why and the how of behavioural operational research An invitation to potential sceptics. *European Journal of Operational Research*, 249, 796–805. doi:10.1016/j.ejor.2015.09.034.

- Bryant, J., Ritchie, C., & Taket, A. (1994). Messages for the OR practitioner. In C. Ritchie, A. Taket, & J. Bryant (Eds.), Community works: 26 case studies showing community operational research in action. Sheffield: Pavic Press.
- Burns, D. (2018). Deepening and scaling participatory research with the poorest and most marginalised. European Journal of Operational Research in press. doi:10.1016/j.ejor.2017.11.025.
- Calvard, T. S. (2016). Big data, organizational learning, and sensemaking: Theorizing interpretive challenges under conditions of dynamic complexity. *Management Learning*, 47(1), 65–82. doi:10.1177/1350507615592113.
- Caragliu, A., Del Bo, C., & Nijkamp, P. (2009). "Smart cities in Europe. In 3rd central European conference in regional science CERS, 2009 (pp. 45–59). http://www.inta-aivn.org/images/cc/Urbanism/background%20documents/01\_03\_Nijkamp.pdf (Accessed 18 August 2016).
- Nijkamp.pdf (Accessed 18 August 2016).
  Caulkins, J. P., Eeelman, E., Ratnatunga, M., & Schaarsmith, D. (2008). Operations research and public policy for Africa: Harnessing the revolution in management science instruction. *International Transactions in Operations Research*, 15(2), 151–171. doi:10.1111/j.1475-3995.2008.00628.x.
- Chen, Y. C., & Ahn, M. (Eds.). (2017). Routledge handbook of information technology in government. New York: Routledge.
- Chen, Y., & Kesten, O. (2016). Chinese college admissions and school choice reforms: A theoretical analysis. *Journal of Political Economy*, 125(1), 99–139. doi:10.1086/689773.
- Cohen, C., & Midgley, G. (1994). The north humberside diversion from custody project for mentally disordered offenders: Research report. Hull: Centre for Systems Studies
- Córdoba, J.-R. (2009). Critical reflection in planning information systems: A contribution from critical systems thinking. *Information Systems Journal*, *19*(2), 123–147. doi:10.1111/j.1365-2575.2007.00284.x.
- Córdoba, J.-R., & Midgley, G. (2003). Addressing organisational and societal concerns: An application of critical systems thinking to information systems planning in Colombia. In J. Cano (Ed.), Critical reflections on information systems: A systemic approach. Hershey, PA: Idea Group.
- Córdoba, J.-R., & Midgley, G. (2006). Broadening the boundaries: An application of critical systems thinking to IS planning in Colombia. *Journal of the Operational Research Society*, 57, 1064–1080. doi:10.1057/palgrave.jors.2602081.
- Córdoba, J.-R., & Midgley, G. (2008). Beyond organisational agendas: Using boundary critique to facilitate the inclusion of societal concerns in information systems planning. European Journal of Information Systems, 17, 125–142. doi:10.1057/ejis. 2008.4.
- Córdoba-Pachón, J.-R. (2010). Systems practice in the information society. London: Routledge.
- Couldry, N., & Powell, A. (2014). Big data from the bottom up. Big Data & Society, 1(2), 20–53. doi:10.1177/2053951714539277.
- Creswell, J. W. (2014). Research design: Qualitative, quantitative and mixed methods approaches (4th Edition). Los Angeles: Sage.
- Davis, M. (2007). Planet of slums. London: Verso.
- Defilippis, J., & Saegert, S. (2012). Community development reader (2nd Edition). New York: Routledge.
- Dirks, S., & Keeling, M. (2009). A vision of smarter cities: How cities can lead the way into a prosperous and sustainable future. Somers NY: IBM.
- Duran, S., Ergun, O., Keskinocak, P., & Swann, J. (2012). Humanitarian logistics: Advanced purchasing and pre-positioning of relief items, *Handbook of global logistics: Transportation in international supply chains, international series in operations research & management science: Vol. 181* (pp. 447–462). Bookbinder, HH: Springer. doi:10.1007/978-1-4419-6132-7\_18.
- Eisenman, D. P., Cordasco, K. M., Asch, S., Golden, J. F., & Glik, D. (2007). Disaster planning and risk communication with vulnerable communities: Lessons from Hurricane Katrina. *American Journal of Public Health*, 97(S1), S109–S115. doi:10. 2105/AJPH.2005.084335.
- Ekici, A., Keskinocak, P., & Swann, J. (2014). Modeling influenza pandemic and planning food distribution. Manufacturing and Service Operations Management, 16(1), 11–27. doi:10.1287/msom.2013.0460.
- Ergun, O., Guyi, L., Heier-Stamm, J., Keskinocak, P., & Swann, J. (2014). Improving humanitarian operations through collaboration. *Production and Operations Man*agement, 23(6), 1002–1014. doi:10.1111/poms.12107.
- Espinosa, A., & Duque, C. (2018). Complexity management and multi-scale governance: A case study in an Amazonian indigenous association. *European Journal* of Operational Research in press. doi:10.1016/j.ejor.2017.07.049.
- Flood, R. L., & Jackson, M. C. (Eds.). (1991). Critical systems thinking: Directed readings. Chichester: Wiley.
- Flood, R. L., & Romm, N. R. A. (Eds.). (1996). Critical systems thinking: Current research and practice. New York: Plenum.
- Foote, J., Ahuriri-Driscoll, A., Hepi, M., Midgley, G., & Earl-Goulet, J. (2016). Systemic evaluation of a community environmental management programme. Research Memorandum 97. Hull: Hull University Business School.
- Foote, J., Baker, V., Gregor, J., Hepi, M., Houston, D., & Midgley, G. (2007). Systemic problem structuring applied to community involvement in water conservation. *Journal of the Operational Research Society*, 58(5), 645–654. doi:10.1057/palgrave. jors.260224
- Foote, J., Hepi, M., Rogers-Koroheke, M., & Taimona, H. (2005). Urban water decision making project: Learning from the stories of Nga Puna Wai o Hokianga. Christchurch: Institute of Environmental Science and Research.
- Foote, J., Hepi, M., Rogers-Koroheke, M., & Taimona, H. (2017). Supporting indigenous environmental health action: A vignette. In J. McIntyre-Mills, N. R. A. Romm, & Y. Corcoran-Nantes (Eds.), Balancing individualism and col-

- lectivism: Social and environmental justice. New York: Springer. doi:10.1007/978-3-319-58014-2 18.
- Franco, L. A., & Hämäläinen, R. P. (2016). Behavioural operational research: Returning to the roots of the OR profession. European Journal of Operational Research, 249, 791–795. doi:10.1016/j.ejor.2015.10.034.
- Friend, J. (2004). Perspectives of engagement in community operational research. In G. Midgley, & A. E. Ochoa-Arias (Eds.), Community operational research: OR and systems thinking for community development. New York: Kluwer. doi:10.1007/ 978-1-4419-8911-6\_9.
- Galea, S., & Vlahov, D. (Eds.). (2005). Handbook of urban health: Populations, methods, and practice. New York: Springer.
- Gomes, S. L., Hermans, L. M., & Thissen, W. A. (2018). Extending community operational research to address institutional aspects of societal problems: Experiences from peri-urban Bangladesh. *European Journal of Operational Research* in press. doi:10.1016/j.ejor.2017.11.007.
- Gregory, A. J., & Atkins, J. P. (2018). Community operational research and citizen science: Two icons in need of each other? *European Journal of Operational Research* in press. doi:10.1016/j.ejor.2017.07.037.
- Gregory, W.J. (1992). Critical systems thinking and pluralism: A new constellation. Ph.D. thesis, City University, London.
- Gregory, W. J. (2000). Transforming self and society: A "critical appreciation" model. Systemic Practice and Action Research, 13(4), 475–501. doi:10.1023/A: 10.09541430809
- Gregory, W. J., & Midgley, G. (2000). Planning for disaster: Developing a counselling service. Journal of the Operational Research Society, 51, 278–290. doi:10.1057/ palgrave.jors.2600912.
- Gregory, W. J., & Romm, N. R. A. (2001). Critical facilitation: Learning through intervention in group processes. *Management Learning*, 32(4), 453–467. doi:10.1177/1350507601324003.
- Gregory, W. J., Romm, N. R. A., & Walsh, M. P. (1994). The Trent quality initiative: A multi-agency evaluation of quality standards in the national health service. Hull: Centre for Systems Studies.
- Gu, J., & Zhu, Z. (2000). Knowing wuli, sensing shili, caring for renli: Methodology of the WSR approach. Systemic Practice and Action Research, 13(1), 11–20. doi:10. 1023/A:1009567421256.
- Hall, P., & Tewdwr-Jones, M. (2010). Urban and regional planning (5th ed). New York: Routledge.
- Hämäläinen, R. P., Luoma, J., & Saarinen, E. (2013). On the importance of behavioral operational research: The case of understanding and communicating about dynamic systems. European Journal of Operational Research, 228, 623–634. doi:10.1016/j.ejor.2013.02.001.
- Helfgott, A. (2018). Operationalizing Systemic Resilience. European Journal of Operational Research in press.
- Hepi, M., Foote, J., Rogers-Koroheke, M., & Taimona, H. (2007). 'Koe wai hoki koe?!' (who the hell are you?!): Issues of trust in cross-cultural collaborative research. Kotuitui: New Zealand Journal of Social Sciences Online, 2, 37–53. doi:10.1080/1177083X.2007.9522423.
- Houston, J. B., Hawthorne, J., Perreault, M. F., Park, E. H., Goldstein Hode, M., Halli-well, M. R., et al. (2015). Social media and disasters: A functional framework for social media use in disaster planning, response, and research. *Disasters*, 39(1), 1–22. doi:10.1111/disa.12092.
- Masterman, J. H., Peacock, W. G., Van Zandt, S. S., Grover, H., Field Schwartz, L., & Cooper, J. T. (2014). Planning for community resilience: A handbook for reducing vulnerability to disasters. Washington, DC: Island Press.
- Hindle, G. A., & Vidgen, R. (2018). Developing a business analytics methodology: A case study in the foodbank sector. European Journal of Operational Research in press. doi:10.1016/j.ejor.2017.06.031.
- INFORMS. (2016a). "Doing good with good OR student paper competition". https://www.informs.org/Recognize-Excellence/INFORMS-Prizes-Awards/ Doing-Good-with-Good-OR-Student-Paper-Competition (Accessed 24 November 2016).
- INFORMS. (2016b). "Pro bono analytics". http://connect.informs.org/ probonoanalytics/home (Accessed 24 November 2016).
- INFORMS (2016c). "Certified analytics professional." https://www.certifiedanalytics.org/ (Accessed 1 January 2017).
- INFORMS (2016d). "Analytics maturity model". https://www.informs.org/ Apply-Operations-Research-and-Analytics/Analytics-Maturity-Model (Accessed 1 January 2017).
- Jackson, M. C. (1985). Social systems theory and practice: The need for a critical approach. *International Journal of General Systems*, 10, 135–151. doi:10.1080/03081078508934877.
- Jackson, M. C. (1987). Community operational research: Purposes, theory and practice. *Dragon*, 2(2), 47–73.
- Jackson, M. C. (1988). Some methodologies for community operational research. Journal of the Operational Research Society, 39(8), 715–724. doi:10.1057/palgrave. jors.0390802.
- Jackson, M. C. (1991). Systems methodology for the management sciences. New York: Plenum.
- Jackson, M. C. (2006). Beyond problem structuring methods: Reinventing the future of OR/MS. *Journal of the Operational Research Society*, 57(7), 868–878. doi:10. 1057/palgrave.jors.2602093.
- Jellie, M., Foote, J., North, N., Rogers, M., Taimona, H., & Marino, M. (2003). A case study of improving drinking water quality in the Hokianga. In *Proceedings of the New Zealand water and wastes association conference*, 17-19 September 2003.
- Johnson, S. K., Hekman, D. R., & Chan, E. T. (2016). If there's only one woman in your candidate pool, there's statistically no chance she'll be hired.

- Harvard Business Review, April 26, 2016. https://hbr.org/2016/04/if-theres-only-one-woman-in-your-candidate-pool-theres-statistically-no-chance-shell-be-hired (Accessed 1 January 2017).
- Johnson, M. P. (2016). An agenda for diversity and inclusion-related research within OR/MS/Analytics. Presented at INFORMS fall national conference, Nashville, TN November 13, 2016 https://works.bepress.com/michael\_johnson/77/ (Accessed 1 lanuary 2017).
- Johnson, M. P. (2015). Data, analytics and community-based organizations: Transforming data to decisions for community development. I/S: A Journal of Law and Policy for the Information Society: Big Data Future Part Two, 11(1), 49–96
- Johnson, MP (Ed.). (2012a). Community-based operations research: Decision modeling for local impact and diverse populations. New York: Springer.
- Johnson, M. P. (2012b). Community-based operations research: Introduction, theory and applications. In M. P. Johnson (Ed.), Community-based operations research: Decision modeling for local impact and diverse populations (pp. 3–36). New York: Springer. doi:10.1007/978-1-4614-0806-2\_1.
- Johnson, M. P., Drew, R. B., Keisler, J., & Turcotte, D. (2012). What is a strategic acquisition? Decision modeling in support of foreclosed housing redevelopment. Socio-Economic Planning Sciences, 46(3), 194–204. doi:10.1016/j.seps.2012.05.002.
- Johnson, M. P., Hollander, J., & Davenport Whiteman, E. (2015). Data and analytics for neighborhood development: Smart shrinkage decision modeling in Baltimore, Maryland. In S. Geertman, J. Ferreira, R. Goodspeed, & J. Stillwell (Eds.), Planning support systems and smart cities (pp. 61–76). Switzerland: Springer. doi:10.1007/978-3-319-18368-8\_4.
  Johnson, M. P., & Jani, S. (2016). "Measuring success: Community analytics for local
- Johnson, M. P., & Jani, S. (2016). "Measuring success: Community analytics for local economic development". Working paper. https://works.bepress.com/michael\_ johnson/70/ (Accessed 2 January 2017).
- Johnson, M. P., Keisler, J. M., Solak, S., Turcotte, D. A., Bayram, A., & Drew, R. B. (2015). Decision science for housing and community development: Localized and evidence-based responses to distressed housing and blighted communities. New York: Wiley.
- Kaplan, E (Ed.). (2015). Editor's cut, volume 4: Confronting public problems with operations research. Catonsville, MD: Institute for Operations Research and the Management Sciences.
- Keisler, J., Turcotte, D. A., Drew, R. B., & Johnson, M. P. (2014). Values structuring and strategy design for housing and community development. EURO Journal on Decision Processes, 2(3-4), 221–256. doi:10.1007/s40070-014-0032-y.
- Keys, P., & Midgley, G. (2002). The process of OR. Journal of the Operational Research Society, 53(2), 123–125. doi:10.1057/palgrave.jors.2601313.
- Kirby, M. W. (2007). Paradigm change in operations research: Thirty years of debate. Operations Research, 55(1), 1–13. doi:10.1287/opre.1060.0310.
- LeClerc, P. D., McLay, L. A., & Mayorga, M. E. (2012). Modeling equity for allocating public resources. In M. P. Johnson (Ed.), Community-based operations research: Decision modeling for local impact and diverse populations (pp. 97–118). New York: Springer. doi:10.1007/978-1-4614-0806-2\_4.
- Levy, J. M. (2017). Contemporary urban planning (11th Edition). New York: Routledge. Li, Y., & Zhu, Z. (2014). Soft OR in China: A critical report. European Journal of Operational Research, 232(3), 427–434. doi:10.1016/j.ejor.2013.04.035.
- Liberatore, F., Ortuño, M. T., Tirado, G., Vitoriano, B., & Scaparra, M. P. (2014). A hierarchical compromise model for the joint optimization of recovery operations and distribution of emergency goods in humanitarian logistics. *Computers & Operations Research*, 42, 3–13. doi:10.1016/j.cor.2012.03.019.
- Liberatore, M. J., & Luo, W. (2010). The analytics movement: Implications for operations research. *Interfaces*, 40(4), 313–324. doi:10.1287/inte.1100.0502.
- Lopez, C. (2015). Modeling sustainability of participatory information systems for urban communities: A mixed-method approach. Pittsburgh: University of Pittsburgh.
- Luoma, J. (2016). Model-based organizational decision making: A behavioral lens. European Journal of Operational Research, 249, 816–826. doi:10.1016/j.ejor.2015. 08.039.
- Manoharan, A. (Ed.). (2015). E-government and websites: A public solutions handbook. New York: Routledge.
- Maposa, D., Cochran, J. J., & Lesaoana, M. (2016). Modelling non-stationary annual maximum flood heights in the lower Limpopo river basin of Mozambique. Jàmbá: Journal of Disaster Risk Studies, 8(1). doi:10.4102/jamba.v8i1.185.
- Martin, S. A. (2015). A framework to understand the relationship between social factors that reduce resilience in cities: Application to the city of Boston. *Inter*national Journal of Disaster Risk Reduction, 12, 53–80. doi:10.1016/j.ijdrr.2014.12. 001.
- McLaren, D., & Agyeman, J. (2015). Sharing cities: A case for truly smart and sustainable cities. Cambridge, MA: MIT Press.
- Metcalfe, M. (2008). Pragmatic inquiry. *Journal of the Operational Research Society*, 59(8), 1091–1099. doi:10.1057/palgrave.jors.2602443.
- Mettler, S. (2011). The submerged state: How invisible government policies undermine American democracy. Chicago: The University of Chicago Press.
- Midgley, G. (1992). Pluralism and the legitimation of systems science. Systems Practice, 5, 147–172. doi:10.1007/BF01059938.
- Midgley, G. (1994). Ecology and the poverty of humanism: A critical systems perspective. Systems Research, 11, 67–76. doi:10.1002/sres.3850110406.
- Midgley, G. (1995). The nature of critical self-reflection. *Journal of the Operational Research Society*, 46(4), 547–552. doi:10.1057/jors.1995.78.
- Midgley, G. (1996). What is this thing called critical systems thinking? In R. L. Flood, & N. R. A. Romm (Eds.), Critical systems thinking: Current research and practice New York: Plenum. doi:10.1007/978-1-4757-9883-8\_7.
- Midgley, G. (1997). Dealing with coercion: Critical systems heuristics and beyond. Systems Practice, 10, 37–57. doi:10.1007/BF02557850.

- Midgley, G. (2000). Systemic intervention: Philosophy, methodology, and practice. New York: Kluwer/Plenum.
- Midgley, G. (2004). Five sketches of post-modernism. *Journal of Organizational Transformation and Social Change*, 1, 47–62. doi:10.1386/jots.1.1.47/0.
- Midgley, G. (2006). Systemic intervention for public health. American Journal of Public Health, 96, 466–472. doi:10.2105/AJPH.2005.067660.
- Midgley, G. (2015). Systemic intervention. In H. Bradbury-Huang (Ed.), *The Sage handbook of action research* (3rd Edition). London: Sage. doi:10.4135/9781473921290.n16.
- Midgley, G., Ahuriri-Driscoll, A., Baker, V., Foote, J., Hepi, M., Taimona, H., et al. (2007). Practitioner identity in systemic intervention: Reflections on the promotion of environmental health through Māori community development. Systems Research and Behavioral Science, 24, 233–247. doi:10.1002/sres.827.
- Midgley, G., Gu, J., & Campbell, D. (2000). Dealing with human relations in Chinese systems practice. Systemic Practice and Action Research, 13(1), 71–96. doi:10.1023/A:1009523606235.
- Midgley, G., Johnson, M. P., & Chichirau, G. (2018). What is community operational research? European Journal of Operational Research in press. doi:10.1016/j.ejor. 2017.08.014.
- Midgley, G., & Milne, A. (1995). Creating employment opportunities for people with mental health problems: A feasibility study for new initiatives. *Journal of the Operational Research Society*, 46(1), 35–42. doi:10.1057/jors.1995.4.
- Midgley, G., Munlo, I., & Brown, M. (1998). The Theory and practice of boundary critique: Developing housing services for older people. *Journal of the Operational Research Society*, 49, 467–478. doi:10.1057/palgrave.jors.2600531.
- Midgley, G., Nicholson, J., & Brennan, R. (2017). Dealing with challenges to methodological pluralism: The paradigm problem, psychological resistance and cultural barriers. *Industrial Marketing Management*, 62, 150-159. doi:10.1016/j. indmarman.2016.08.008.
- Midgley, G., & Ochoa-Arias, A. E. (1999). Visions of community for community OR. *Omega*, 27, 259–274. doi:10.1016/S0305-0483(98)00044-9.
- Midgley, G., & Ochoa-Arias, A. E. (2004a). An introduction to community operational research. In G. Midgley, & A. E. Ochoa-Arias (Eds.), Community operational research: OR and systems thinking for community development. New York: Kluwer. doi:10.1007/978-1-4419-8911-6\_1.
- Midgley, G., & Ochoa-Arias, A. E. (Eds.). (2004b). Community operational research: OR and systems thinking for community development. New York: Kluwer.
- Midgley, G., & Pinzón, L. (2011). The implications of boundary critique for conflict prevention. *Journal of the Operational Research Society*, 62(8), 1543–1554. doi:10. 1057/jors.2010.76.
- Midgley, G., & Pinzón, L. (2013). Systemic mediation: Moral reasoning and boundaries of concern. Systems Research and Behavioral Science, 30, 607–632. doi:10.1003/eps.2328
- Midgley, G., & Reynolds, M. (2001). Operational research and environmental management: A new agenda. Birmingham: Operational Research Society.
- Midgley, G., & Reynolds, M. (2004a). Systems/operational research and sustainable development: Towards a new agenda. Sustainable Development, 12, 56–64. doi:10.1002/sd.218.
- Midgley, G., & Reynolds, M. (2004b). Community and environmental OR: Towards a new agenda. In G. Midgley, & A. E. Ochoa-Arias (Eds.), *Community operational research: OR and systems thinking for community development*. New York: Kluwer. doi:10.1007/978-1-4419-8911-6\_13.
- Midgley, G., & Shen, C.-Y. (2007). Toward a Buddhist systems methodology 2: An exploratory, questioning approach. Systemic Practice and Action Research, 20, 195–210. doi:10.1007/s11213-006-9059-8.
- Midgley, G., & Wilby, J. (2000). Systems practice in China: New developments and cross-cultural collaborations. Systemic Practice and Action Research, 13(1), 3–9. doi:10.1023/A:1009515404417.
- Midgley, G., & Wilby, J. (Eds.). (1995). Systems methodology: Possibilities for cross-cultural learning and integration. Hull: Centre for Systems Studies.
- Mingers, J. (2011a). Ethics and OR: Operationalising discourse ethics. European Journal of Operational Research, 210, 114–124. doi:10.1016/j.ejor.2010.11.003.
- Mingers, J. (2011b). Soft OR comes of age but not everywhere!. *Omega*, 39, 729–741. doi:10.1016/j.omega.2011.01.005.
- Mingers, J. (2011c). Taming hard problems with soft O.R. OR/MS Today, 36(2), 48–53.Mingers, J., & Gill, A. (Eds.). (1997). Multimethodology: The theory and practice of combining management science methodologies. Chichester: Wiley.
- Mingers, J., & Rosenhead, J. (2004). Problem structuring methods in action. European Journal of Operational Research, 152, 530-554. doi:10.1016/S0377-2217(03) 00055 0
- Morgan, T. K. K. B., & Fa'aui, T. N. (2018). Empowering indigenous voices in disaster response: Applying the mauri model to New Zealand's worst environmental maritime disaster. European Journal of Operational Research in press. doi:10.1016/j.ejor.2017.05.030.
- Mortenson, M. J., Doherty, N. F., & Robinson, S. (2014). Operational research from Taylorism to terabytes: A research agenda for the analytics age. European Journal of Operational Research, 241(3), 583–595. doi:10.1016/j.ejor.2014.08.029.
- Munday, P. G. (2015). Developing a systems approach for multi-agency co-ordination and community engagement in disaster recovery. PhD Thesis, University of Hull, Hull
- Murthy, P. N. (1994). Inquiry systems of upanishads. Systems Practice, 7(4), 457–463. doi:10.1007/BF02169366.
- Mwiti, F., & Goulding, C. (2018). Gender and strategies for community improvement: An ethnography of community based organizations ('Chamas') and women's interventions in the Nairobi slums. European Journal of Operational Research in press. doi:10.1016/j.ejor.2017.12.009.

- National Federation for Catholic Youth Ministry (2008). "Social Justice". Washington, D.C.: National Catholic Committee for Girl Scouts USA and Camp Fire USA. http://www.nccgscf.org/resources/justice.htm (Accessed 16 August 2016).
- Nussbaumer Knaflic, C. (2015). Storytelling with data: A data visualization guide for business professionals. New York: Wiley.
- Ochoa-Arias, A. E. (1994). The possibilities of community OR in a third world country. *International Transactions in Operational Research*, 1, 345–352. doi:10.1016/0969-6016(94)90034-5.
- Ochoa-Arias, A. E. (2004). An interpretive systemic exploration of community action in Venezuela. In G. Midgley, & A. E. Ochoa-Arias (Eds.), Community operational research: OR and systems thinking for community development. New York: Kluwer. doi:10.1007/978-1-4419-8911-6 12.
- Operational Research Society (2016). "Pro bono O.R. in the third sector". http://www.theorsociety.com/Pages/Probono/Probono.aspx (Accessed 30 December 2016).
- Ormerod, R. (2006). The history and ideas of pragmatism. *Journal of the Operational Research Society*, 57(8), 892–909. doi:10.1057/palgrave.jors.2602065.
- Ormerod, R. J., & Ulrich, W. (2013). Operational research and ethics: A literature review. European Journal of Operational Research, 228(2), 291–307. doi:10.1016/j.eior.2012.11.048.
- Ormerod, R. J. (2014). The mangle of OR practice: Towards more informative case studies of 'technical' projects. *Journal of the Operational Research Society*, 65(8), 1245–1260. doi:10.1057/jors.2013.78.
- Pimentel, B. S., Santibañez Gonzalez, E., & Barbosa, G. N. O. (2016). Decision-support models for sustainable mining networks: Fundamentals and challenges. *Journal of Cleaner Production*, 112, 2145–2157. doi:10.1016/j.jclepro.2015.09.023.
- Pindar, S. (1994). Planning a network response to racial harassment. In C. Ritchie, A. Taket, & J. Bryant (Eds.), Community works: 26 case studies showing community operational research in action. Sheffield: Pavic Press.
- Pinzón-Salcedo, L. A., & Torres-Cuello, M. A. (2018). Community operations research: Developing a systemic peace education programme involving urban and rural communities in Colombia. *European Journal of Operational Research* in press. doi:10.1016/j.ejor.2017.11.040.
- Pollock, S. M., Rothkopf, M. H., & Barnett, A. (Eds.). (1994). *Operations research and the public sector*. Amsterdam: North-Holland.
- Ranyard, J. C., Fildes, R., & Hu, T.-I. (2015). Reassessing the scope of OR practice: The influences of problem structuring methods and the analytics movement. European Journal of Operational Research, 245(1), 1–13. doi:10.1016/j.ejor.2015.01. 058.
- Raymaker, D. M. (2016). Intersections of critical systems thinking and community based participatory research: A learning organization example with the autistic community. Systemic Practice and Action Research, 29(5), 405–423. doi:10.1007/ s11213-016-9376-5.
- Ritchie, C. (2004). Housing in the Dearne valley: Doing community OR with the Thurnscoe tenants housing co-operative. In G. Midgley, & A. E. Ochoa-Arias (Eds.), Community operational research: OR and systems thinking for community development. New York: Kluwer. doi:10.1007/978-1-4419-8911-6\_6.
- Ritchie, C., Taket, A., & Bryant, J. (Eds.). (1994). Community works: 26 case studies showing community operational research in action. Sheffield: Pavic Press.
- Rittel, H. W. J., & Webber, M. M. (1973). Dilemmas in a general theory of planning. *Policy Sciences*, 4(2), 155–169. doi:10.1007/BF01405730.
- Robinson, M. (2016). What is social justice?. Boone NC: Appalachian State University <a href="http://gjs.appstate.edu/social-justice-and-human-rights/what-social-justice-accessed-16">http://gjs.appstate.edu/social-justice-and-human-rights/what-social-justice-accessed-16</a> August 2016].
- Romm, N. R. A. (2018). Reflections on a multi-layered intervention in the South African public education system: Some ethical implications for community operational research. European Journal of Operational Research in press. doi:10.1016/ i.eior.2017.02.022.
- Rosenhead, J. (1986). Custom and practice. Journal of the Operational Research Society, 37(4), 335–343. doi:10.1057/jors.1986.61.
- Rosenhead, J. (1993). Enabling analysis: Across the developmental divide. *Systems Practice*, 6(2), 117–138. doi:10.1007/BF01062247.
- Rosenhead, J., & Mingers, J. (2001). Rational analysis for a problematic world revisited: Problem structuring methods for complexity, uncertainty and conflict (2nd edition). Chichester: Wiley.
- Savas, E. (1969). Simulation and cost-effectiveness analysis of New York's emergency ambulance service. *Management Science*, 15, B608–B627. doi:10.1287/mnsc.15.12. B608.
- Savas, E. (1978). On equity in providing public services. *Management Science*, 24, 800–808. doi:10.1287/mnsc.24.8.800.
- Schafer, J. G., & Gallemore, C. T. (2016). Biases in multicriteria decision analysis: The case of environmental planning in southern Nevada. *Environment and Planning C: Government and Policy*, 34(8), 1652–1675. doi:10.1177/0263774X16629675.
- Scott, R. J., Cavana, R. Y., & Cameron, D. (2016). Recent evidence on the effectiveness of group model building. European Journal of Operational Research, 249, 908–918. doi:10.1016/j.ejor.2015.06.078.
- Shen, C.-Y., & Midgley, G. (2007a). Toward a Buddhist systems methodology 3: An application in a Taiwanese non-governmental organization. *Systemic Practice and Action Research*, 20, 211–244. doi:10.1007/s11213-006-9060-2.
- Shen, C.-Y., & Midgley, G. (2007b). Toward a Buddhist systems methodology 1: Comparisons between Buddhism and systems theory. *Systemic Practice and Action Research*, 20, 167–194. doi:10.1007/s11213-006-9058-9.
- Shen, C.-Y., & Midgley, G. (2015). Action research in a problem avoiding culture using a Buddhist systems methodology. *Action Research*, 13(2), 170–193. doi:10. 1177/1476750314558428.

- Shi, P. (2015). Guiding school-choice reform through novel applications of operations research. *Interfaces*, 45(2), 117–132. doi:10.1287/inte.2014.0781.
- Siegel, E. (2016). Predictive analytics: The power to predict who will click, buy, lie, or die (2nd Edition). New York: Wiley.
- Simchi-Levi, D. (2006). The state of operations research. *OR/MS Today*, 33(1), 52–55. Simchi-Levi, D. (2009). Not the appropriate outlet. *OR/MS Today*, 36(2), 21.
- Smith, L. T. (1999). Decolonizing methodologies: Research and indigenous peoples. London: Zed Books.
- Souza, R., Rosenhead, J., Salhofer, S. P., Valle, R. A. B., & Lins, M. P. E. (2014). Definition of sustainability impact categories based on stakeholder perspectives. *Journal of Cleaner Production*, 105, 41–51. doi:10.1016/j.jclepro.2014.09.051.
- Sova, C. A., Helfgott, A., Chaudhury, A. S., Matthews, D., Thornton, T. F., & Vermeulen, S. J. (2015). Multi-level stakeholder influence mapping: Visualizing power relations across actor levels in Nepal's agricultural climate change adaptation regime. Systemic Practice and Action Research, 28, 383–409. doi:10.1007/s11213-014-9335-y.
- Spaans, M., & Waterhout, B. (2017). Building up resilience in cities worldwide Rotterdam as participant in the 100 resilient cities programme. Cities, 61, 109–116. doi:10.1016/j.cities.2016.05.011.
- Stokey, E., & Zeckhauser, R. (1978). A primer for policy analysis. New York: W.W. Norton & Company.
- Taket, A. (1994). Undercover agency? Ethics, responsibility and the practice of OR. Journal of the Operational Research Society, 45(2), 123–132. doi:10.1057/jors.1994.
- Taket, A., & White, L. (1993). After OR: An agenda for postmodernism and poststructuralism in OR. Journal of the Operational Research Society, 44, 867–881. doi:10.1038/sj/jors/0440902.
- Taket, A., & White, L. (2000). Partnership and participation: Decision-making in the multiagency setting. Chichester: Wiley.
- Tan, B., Watson, R., & Wei, K. K. (1995). National culture and group support systems: Filtering communication to dampen power differentials. *European Journal of Information Systems*, 4(2), 82–92. doi:10.1057/ejis.1995.10.
- Tavella, E., & Franco, L. A. (2015). Dynamics of group knowledge production in facilitated modelling workshops: An exploratory study. Group Decision and Negotiation, 24(3), 451–475. doi:10.1007/s10726-014-9398-2.
- The Movement for Black Lives (2016). "Platform". https://policy.m4bl.org/platform/ (Accessed 1 January 2017).
- Thompson, J. P., Howick, S., & Belton, V. (2016). Critical learning incidents in system dynamics modelling engagements. European Journal of Operational Research, 249, 945–958. doi:10.1016/j.ejor.2015.09.048.
- Thunhurst, C. P. (2013). Public health systems analysis where the River Kabul meets the River Indus. *Globalization and Health*, 9(1), 39. doi:10.1186/1744-8603-9-39.
- Tirivanhu, P., Matondi, P. B., & Sun, D. (2016). Systemic evaluation of a comprehensive community initiative based on boundary critique in Mhakwe ward in Zimbabwe. Systemic Practice and Action Research, 29(6), 541–564. doi:10.1007/s11213-016-9377-4.
- Turcotte, D. A., Johnson, M. P., Chaves, E. J., Drew, R. B., & Sullivan, F. M. (2015). Reconstructing neighborhoods: Two case studies in foreclosed housing acquisition and redevelopment by community development corporations in Massachusetts. Housing and Society, 42(1), 17–39. doi:10.1080/08882746.2015.1020700.
- Ufua, D. E., Papadopoulos, T., & Midgley, G. (2018). Systemic lean intervention: Enhancing lean with community operational research. *European Journal of Operational Research* in press. doi:10.1016/j.ejor.2017.08.004.
- Ulrich, W. (1987). Critical heuristics of social systems design. European Journal of Operational Research, 31, 276–283. doi:10.1016/0377-2217(87)90036-1.
- Ulrich, W. (1993). Some difficulties of ecological thinking, considered from a critical systems perspective: A plea for critical holism. Systems Practice, 6, 583–611. doi:10.1007/BF01059480.
- Ulrich, W. (1994). Critical heuristics of social planning: A new approach to practical philosophy. Chichester: Wiley.
- Ulrich, W. (1996). A primer to critical systems heuristics for action researchers. Hull: Centre for Systems Studies.
- Urban Research Based Action Network (2016). "Home". http://urbanresearchnet work.org/ (Accessed 1 January 2017).
- Velez-Castiblanco, J., Brocklesby, J., & Midgley, G. (2016). Boundary games: How teams of OR practitioners explore the boundaries of intervention. European Journal of Operational Research, 249, 968–982. doi:10.1016/j.ejor.2015.08.006.
- Vertovec, S., & Wessendorf, S. (Eds.). (2010). Multiculturalism backlash: European discourses, policies and practices. Abington, UK: Routledge.
- Walsh, M., Kittler, M. G., & Mahal, D. (2018). Towards a new paradigm of health-care: Addressing challenges to professional identities through community operational research. European Journal of Operational Research in press. doi:10.1016/j.ejor.2017.05.052.
- Walters, P. (2015). The problem of community resilience in two flooded cities: Dhaka 1998 and Brisbane 2011. Habitat International, 50, 51-56. doi:10.1016/j. habitatint.2015.08.004.
- Waltner-Toews, D., Kay, J., Murray, T. P., & Neudoerffer, C. (2004). Adaptive methodology for ecosystem sustainability and health (AMESH): An introduction. In G. Midgley, & A. E. Ochoa-Arias (Eds.), Community operational research: OR and systems thinking for community development. New York: Kluwer. doi:10.1007/978-1-4419-8911-6\_14.
- Wang, Z. (2000). Meta-decision making: Concepts and paradigm. Systemic Practice and Action Research, 13(1), 111–115. doi:10.1023/A:1009527707143.
- Weaver, R., Bagchi-Sen, S., Knight, J., & Frazier, A. E. (2017). Shrinking cities: Understanding urban decline in the united states. New York: Routledge.

- Wenstop, F., & Koppang, H. (2009). On operations research and value conflicts. *Omega*, 37, 1109–1120. doi:10.1016/j.omega.2008.10.001.
- White, A., & Trump, K.-S. (2016). The promises and pitfalls of 311 data. *Urban Affairs Review*. doi:10.1177/1078087416673202.
- White, L. A. (1994). Development options for a rural community in Belize Alternative development and operational research. *International Transactions in Operational Research*, 1(4), 453–462. doi:10.1111/j.1475-3995.1994.00322.x.
- White, L., Burger, K., & Yearworth, M. (2016). Understanding behaviour in problem structuring methods interventions with activity theory. *European Journal of Operational Research*, 249, 983–1004. doi:10.1016/j.ejor.2015.07.044.
- White, L., Smith, H., & Currie, C. (2011). OR in developing countries: A review. European Journal of Operational Research, 208(1), 1–11. doi:10.1016/j.ejor.2010.02.015.
- Winston, W. L., & Albright, S. C. (2016). Practical management science (5th Edition). Boston: Cengage Learning.
- Xing, Y., Li, L., Bi, Z., Wilamowska-Korsak, M., & Zhang, L. (2013). Operations research (OR) in service industries: A comprehensive review. *Systems Research and Behavioral Science*, 30(3), 300–353. doi:10.1002/sres.2185.
- Yearworth, M., & White, L. (2018). Spontaneous emergence of community OR: Self-initiating, self-organising problem structuring mediated by social media. European Journal of Operational Research in press. doi:10.1016/j.ejor.2018.01.024.
- Zhu, Z. (2000). Dealing with a differentiated whole: The philosophy of the WSR approach. Systemic Practice and Action Research, 13(1), 21-57. doi:10.1023/A: 1009519505326.
- Zhong, Q., Karner, A., Kuby, M., & Golub, A. (2017). A multiobjective optimization model for locating affordable housing investments while maximizing accessibility to jobs by public transportation. Environment and Planning B: Urban Analytics and City Science in press. doi:10.1177/2399808317719708.

سايت مرجع مقالات open access سايت مرجع مقالات